

E.ON Energy Research Center FCN | Institute for Future Energy Consumer Needs and Behavior

FCN Working Paper No. 9/2013

# Modeling the Diffusion of Residential Photovoltaic Systems in Italy: An Agent-based Simulation

Johannes Palmer, Giovanni Sorda and Reinhard Madlener

May 2013

# Institute for Future Energy Consumer Needs and Behavior (FCN)

School of Business and Economics / E.ON ERC





#### FCN Working Paper No. 9/2013

#### Modeling the Diffusion of Residential Photovoltaic Systems in Italy: An Agent-based Simulation

May 2013

Authors' addresses:

Johannes Palmer Institute of Heat and Mass Transfer (WSA) RWTH Aachen University Eilfschornsteinstrasse 18 52062 Aachen, Germany E-mail: palmer@wsa.rwth-aachen.de

Giovanni Sorda, Reinhard Madlener Institute for Future Energy Consumer Needs and Behavior (FCN) School of Business and Economics / E.ON Energy Research Center RWTH Aachen University Mathieustrasse 10 52074 Aachen, Germany E-mail: GSorda@eonerc.rwth-aachen.de, RMadlener@eonerc.rwth-aachen.de

Publisher: Prof. Dr. Reinhard Madlener Chair of Energy Economics and Management Director, Institute for Future Energy Consumer Needs and Behavior (FCN) E.ON Energy Research Center (E.ON ERC) RWTH Aachen University Mathieustrasse 10, 52074 Aachen, Germany Phone: +49 (0) 241-80 49820 Fax: +49 (0) 241-80 49829 Web: www.eonerc.rwth-aachen.de/fcn E-mail: post\_fcn@eonerc.rwth-aachen.de

# Modeling the diffusion of residential photovoltaic systems in Italy: An agent-based simulation

Johannes Palmer<sup>1,\*</sup>, Giovanni Sorda<sup>2</sup>, and Reinhard Madlener<sup>2</sup>

<sup>1</sup>Institute of Heat and Mass Transfer (WSA), RWTH Aachen University Eilfschornsteinstr. 18, 52062 Aachen, Germany

<sup>2</sup>Institute for Future Energy Consumer Needs and Behavior (FCN)

School of Business and Economics / E.ON Energy Research Center

RWTH Aachen University, Mathieustr. 10

52074 Aachen, Germany

## May 2013

#### Abstract

We propose an agent-based model to simulate the diffusion of small PV systems among single- or two-family homes in Italy over the 2006-2026 period. To this end, we explicitly model the geographical distribution of the agents in order to account for regional differences across the country. The adoption decision is assumed to be influenced predominantly by (1) the payback period of the investment, (2) its environmental benefit, (3) the household's income, and (4) the influence of communication with other agents. For the estimation of the payback period, the model considers investment costs, local irradiation levels, governmental support, earnings from using self-produced electricity vs. buying electricity from the grid, as well as various administrative fees and maintenance costs. The environmental benefit is estimated by a proxy for the  $CO_2$  emissions saved. The level of the household income is associated with the specific economic conditions of the region where the agent is located, as well as the agent's socio-economic group (age group, level of education, household type). Finally, the influence of communication is measured by the number of links with other households that have already adopted a PV system. In each simulation step, the program dynamically updates the social system and the communication network, while the evolution of the PV system's investment costs depend on a one-factor experience curve model that is based on the exogeneous development of the global installed PV capacity. Our results show that Italy's domestic PV installations are already beyond an initial stage of rapid growth and, though likely to spread further, they will do so at a significantly slower rate of diffusion.

Keywords: PV, Technological diffusion, Agent-based modeling, Italy

<sup>\*</sup>Corresponding author. Tel.: +49-241-80 94 491. E-mail: palmer@wsa.rwth-aachen.de

## 1 Introduction

Following the introduction of a governmental incentive program, the Italian photovoltaics (PV) market has experienced a remarkable growth. Electricity generated by PV systems increased from 35 GWh in 2006 to 10,796 GWh in 2011, an astounding increment (GSE, 2012a; see also Figure 1 and Table 1). Italy has thus become one of the world's leading PV markets, accounting for about 18% of the global installed PV capacity in 2011 (EPIA, 2012).

Nevertheless, the diffusion of PV across Italy has followed a rather peculiar pattern. The number of installed PV systems is much higher in the north, although the irradiation level is lower there compared to other regions of the country. In addition, most of the installed systems in the north belong to private households and are thus characterized by a small rated power. However, while small-scale PV systems up to 20 kW are overwhelming in number (88% of the total, as of 2011), they account for only 15.5% of Italy's installed PV power (GSE, 2012c, , see also Figure 1). Furthermore, the share of small PV systems with respect to installed capacity has fallen steadily (from 66% in 2006 to 15.5% in 2011) due to the more recent installation of large PV farms (mostly located in Central and Southern Italy), a trend that strongly contributed to the PV boom in Italy (GSE, 2012c). As a result, the number, size and electricity generation of PV systems in Italy are rather unevenly spread across the country.

It is thus relevant to investigate whether the residential PV market will grow further, or whether the Italian PV market will be dominated in the future by large PV farms. The objective of this article is to simulate the future diffusion of small residential PV systems under different conditions. Due to the multitude of factors influencing a household's investment decision in favor of an innovative energy technology such as PV, we designed and implemented an agent-based simulation model (ABM). ABMs provide a suitable framework to explicitly model the adoption decision process of the members (agents) of a heterogeneous social system based on their individual preferences, behavioral rules, and interactions/communications within a social network.

We explicitly model the geographical distribution of the agents in order to account for the regional differences that have strongly influenced the PV diffusion in Italy. The investment in a PV system is assumed to depend mainly on (1) the payback period, (2) the environmental benefit of the investment, (3) the household's income, and (4) the influence of communication with other agents. For the estimation of the payback period, the model considers investment costs, local irradiation levels, feed-in tariffs, earnings from using selfgenerated electricity vs. buying electricity from the grid, as well as various administrative fees and maintenance costs. The environmental benefit of the PV system is estimated via a proxy for the amount of  $CO_2$  saved. The level of the household income is associated with the specific economic conditions of the region where the agent is located, as well



Figure 1: Evolution of Italy's PV market, 2006–2011

Source: GSE (2012c)

Table 1: Evolution of Italy's electricity generation, 2006–2011

		2006	2007	2008	2009	2010	2011
Total electricity generation thereof PV el. generation	[TWh] [GWh]	$352.7 \\ 35$	$354.5 \\ 39$	$353.6 \\ 193$	333.3 676	$342.9 \\ 1,906$	346.4 10,796
Share of RES in total electricity generation	[%]	15.9	16.0	16.6	18.8	20.1	23.5
Share of PV in total electricity generation	[%]	0.01	0.01	0.05	0.20	0.56	3.12

Source: GSE (2012a)

as the agent's socio-economic group (age group, level of education, and household type). Finally, the influence of communication is measured by the number of links with other households that have already adopted a PV system. It is assumed that each adopter communicates predominantly, but not uniquely, with other households that belong to the same socio-economic group. Furthermore, the likelihood that different groups interact with each other varies across the categories of agents considered.

Following Schwarz and Ernst (2009), an important contribution to the current literature on PV adoption and diffusion simulations is the inclusion of adaptive socio-economic categories to represent heterogeneous household groups with distinctive attitudes toward adoptions and innovations. The socio-economic groups considered here are based on the Sinus-Milieus<sup>®</sup> categorization developed by the Sinus-Institut (2011).<sup>1</sup> In particular, the

<sup>&</sup>lt;sup>1</sup>The Sinus-Milieus<sup>®</sup> are a registered commercial product of a marketing company that does not disclose the rules or the questionnaires used to generate these socio-economic characterizations. We

Conto Energia	Issue	Cap on cumulative PV installed capacity	Reason for update
$     \begin{array}{c}       1 \\       2 \\       3 \\       4 \\       5     \end{array} $	08/2005 04/2007 01/2011 06/2011 09/2012	100 MW, updated to 500 MW by 2015 1,200 MW, updated to 3000 MW by 2015 8,000 MW by 2020 23,000 MW by 2016, registration required Max + 3,000 MW/a, registration required	Adjustments Adjustments Cap reached Cap almost reached Still in place
Courses 1	10E (900F	9007 $9010$ $9011$ $9019$	

Table 2: Date of issue, support cap, and reasons for revision of the Conto Energia 1–5

Source: MSE (2005, 2007, 2010, 2011, 2012)

Sinus-Milieu<sup>®</sup> paradigm is most relevant for the distribution of the households' income and the determination of group-specific social communication networks. In each simulation step, the social system and the communication network are updated dynamically in order to account for demographic changes and new adopters among the population of agents.

The remainder of the paper is structured as follows. Section 2 provides a brief introduction to the current Italian PV support policy. Section 3 gives an overview of the relevant literature concerning the adoption of new technologies, its modeling via agent-based simulation frameworks, and the inclusion of a social system in the modeling architecture. Section 4 presents in detail the structure of the ABM. Section 5 describes the model's calibration, while section 6 discusses the policy scenarios and the simulation results. Finally, section 7 delivers the conclusions of the article and highlights strengths and weaknesses of our analysis.

## 2 The Italian support scheme for PV systems

The current legal framework for the support of PV systems in Italy is called "Conto Energia" (CE). The first CE has been issued in August 2005. Since then, the incentive scheme has been renewed five times with a series of adjustments and changes. An important characteristic of the CE is that support is granted up to a given amount of total installed PV power, as shown in Table 2 (MSE, 2005, 2007, 2010, 2011, 2012).<sup>2</sup>

Each CE guarantees contracts with fixed conditions for 20 years for grid-connected PV systems with at least 1 kW of peak power. Local electricity providers are required by law to buy the electricity that is generated by PV systems. Furthermore, governmental incentives are tax-free. Beginning with CE 2, the government has also reduced the purchase tax from 20% to 10%.

discuss this aspect in more detail in the conclusion (see section 7).

<sup>&</sup>lt;sup>2</sup>Note that in our model we do not account for the PV installation caps, as we consider only a sub-group of potential adopters and PV systems.



Figure 2: Stages of the Conto Energia by installed PV installed capacity, 2007–2013

Source: Own illustration, based on GSE (2013b)

Table 3: Incentives paid by the Conto Energia, 2006–2012

Year	2006	2007	2008	2009	2010	2011	$2012^{\rm a}$
Incentives [million $\in$ /a] Total incentives [million $\in$ ]	1 1	19 20	91 111	$\begin{array}{c} 304 \\ 415 \end{array}$	$743 \\ 1,158$	$3,835 \\ 4,993$	4,565 9,558

 $^{\rm a}$  Jan.–Sept. 2012

Source: GSE (2012c)

The CE considers two different support schemes. The first scheme is a net metering plan ("scambio sul posto") designed for small PV systems.<sup>3</sup> The plan is meant to favor the direct use of self-produced electricity. Besides a payment for each produced kWh of electricity, the consumer receives additional rewards for directly consuming the self-generated energy. With the introduction of CE 4, direct consumption is rewarded financially, whereas before 2011 consumers received an energy credit. Importantly, energy that is fed into the grid is bought by the local electricity provider at conditions that are less advantageous than direct self-consumption.

The second support scheme is available to all PV systems, but it is designed for larger plants with no or limited direct electricity self-consumption. The electricity produced is sold to the local energy supplier, for which the CE guarantees an additional feed-in payment.

In general, the incentives granted are higher for small PV systems. The Feed-in Tariffs (FiT) increase further for PV systems that are based on innovative technologies or systems that are integrated into the building. Additional payments or bonuses can also

 $<sup>^{3}</sup>$ The first two versions of the CE limited the maximum peak power for this plan to 20 kW. Beginning with CE 3, systems up to 200 kW are also accepted.

be received in the following cases<sup>4</sup>: the adopter owns an energy-saving home; the adopter renews his/her roofs because of asbestos; the adopter lives in a small village with up to 5,000 inhabitants; the PV system was produced in Europe; the PV system is located on a municipal building, in an old industrial area, or in an old garbage dump.

It is important to mention that in each new version of the CE, the FiT were decreased. Since CE 1 was first issued in 2005, the basic support level has been curtailed from approximately 0.45  $\in$ /kWh in 2006 to 0.20  $\in$ /kWh in 2012.<sup>5</sup> Besides a reduction in tax revenues due to cuts in the PV purchase tax and the expenses associated with administrative tasks, the Italian government has spent  $\in$ 9,558 million for PV incentives from 2006 to September 2012 (see Table 3). Due to the high costs, Italy introduced a register for new PV systems with the implementation of CE 4. The register is meant to put a cap on the amount of support granted to PV systems for each year, whereby small PV systems (< 20 kW) still enjoy register priority. Similarly, the latest version of the support scheme (i.e. CE 5) aims at quickly decreasing the level of the feed-in payments, since grid parity was reached around 2011 and the costs of the support program are high. The FiT in CE 5 are set to decrease further by approximately 10% every 6 months for 2.5 years, starting in September 2012. Afterwards, the FiT will be reduced every 6 months by 15%. Figure 2 and Table 3 show the different stages of the CE with respect to the installed PV capacity and the incentives paid per year.

## 3 Literature overview

The modeling and forecasting of technology diffusion has been the focus of theoretical and empirical research since the works of Fourt and Woodlock (1960), Mansfield (1961), Rogers (1962), Chow (1967), and Bass (1969). The adoption and diffusion of innovations is determined by four core elements: the characteristics of the innovation, the structure of the social system where the adoption and diffusion takes place, the communication channels within the social system, and the time-frame of the innovation-decision process (Rogers, 2003). A variety of models focusing on one or more of these elements have been applied to a multitude of research fields and technologies. For an overview, see for instance Mahajan et al. (2000) and Meade and Islam (2006).

In recent years, agent-based simulation models (ABM) have been widely used to simulate the inherent complexity of the adoption and diffusion process (Dawid, 2006; Kiesling et al., 2012). In particular, ABM frameworks replicate the micro-based behavior of economic actors in order to evaluate and explain meso- and macro-level phenomena. They enable modelers to ascribe specific characteristics to the agents, who independently

<sup>&</sup>lt;sup>4</sup>The individual bonuses lead to an increase in the FiT that may range from 5% to 30%. Note that the requirements for the award of a bonus have changed over time (MSE, 2005, 2007, 2010, 2011, 2012).

<sup>&</sup>lt;sup>5</sup>Here we are referring to the basic FiT for small roof-top PV systems.

interact within their environment and among each other according to determined rules (Bonabeau, 2002).

ABM have also been applied to investigate the adoption of various energy technologies (e.g., Schwoon, 2006; Cantono and Silverberg, 2009; Faber et al., 2010; Zhang and Nuttall, 2011; Zhang et al., 2011; Sorda et al., 2013). However, to the best of our knowledge, in the recent literature only Zhao et al. (2011) implement an ABM to simulate the diffusion of PV systems. They evaluate the impact of different governmental incentives, including the impact of investment credit taxes and feed-in tariffs, on the PV diffusion process in two regions in the US.

Nevertheless, the factors influencing PV adoption and their modeling have been the subject of several publications. These can be grouped into three categories: survey-based analyses (Jager, 2006; Faiers and Neame, 2005; Faiers et al., 2007; Yuan et al., 2011; Zhai and Williams, 2012), PV diffusion and forecasting models other than ABM (Guidolin and Mortarino, 2010; Gallo and De Bonis, 2013), and PV grid parity studies (Ayompe et al., 2010; Yang, 2010; Breyer and Gerlach, 2013).

One may suspect that the fast-decreasing installation costs of PV systems and the prospect of grid parity PV electricity generation provide strong incentives for the investment in photovoltaic technology by homeowners. However, the adoption decision is still strongly influenced by the perceived attributes of the innovation, such as installation costs, maintenance, complexity, and environmental concerns (Zhai and Williams, 2012). In addition, the adopter characteristics<sup>6</sup> (Faiers and Neame, 2005; Faiers et al., 2007) as well as the communication network play an important role in the actual diffusion process (Jager, 2006).

In our model, we try to incorporate these three considerations: the specific attributes of the PV technology, the attitudes and preferences of the adopters according to their respective socio-economic groups, as well as the influence of communication among agents.

## 4 Model description

In our model, we consider small grid-connected PV systems in the 1–20 kW range powered by crystalline silicon solar cells (silicon solar cells had a 93% share of the Italian market in 2011; GSE, 2012b). Furthermore, it is assumed that the PV systems are installed on the roofs of single- or two-family houses<sup>7</sup>. The ABM framework simultaneously accounts for the attributes of the PV systems, the attitude of specific adopter groups, and the

 $<sup>^{6}</sup>$ In their study of the adoption of residential PV systems in the UK, Faiers and Neame (2005) and Faiers et al. (2007) based their questionnaires on Rogers' (2003) adopter categories with respect to innovativeness and product characteristics.

<sup>&</sup>lt;sup>7</sup>Small PV systems could also be installed on the roofs of larger multi-flat building blocks. However, the adoption decision would become much more complicated to replicate, as often several house-owners or groups of families cooperatively decide to make a PV investment.



Figure 3: Schematic diagram of the step-wise simulation process

Source: Own illustration

communication network thanks to a multi-attribute utility function (Zhao et al., 2011) weighted by adopter preferences according to different socio-economic classes (Schwarz and Ernst, 2009).

The ABM has been programmed in MATLAB and simulates the PV diffusion process on a step-wise yearly basis. Two key components constitute the core structure of the framework: the agent's adoption decision and the representation of Italy's social system. The decision to invest in a PV system depends on static functions fed with data that, in some cases, account for changes in the underlying social structure and communication network. As a result, specific model parameters are updated after each simulation step, as highlighted in Figure 3. Next, we present in more detail the formulation of the agent's behavioral rules (section 4.1), the modeling of socio-economic attributes in Italy (section 4.2), and the agent's communication network (section 4.3).

## 4.1 Agent's adoption behavior

An agent represents a household living in a single- or two-family house. The decision to invest in a PV system takes place when the utility of the potential adopter surpasses a certain threshold level. The threshold is determined by comparing the simulation results with the actual diffusion of the PV system during the calibration of the model (see section 5 for more details). The utility of agent j equals the sum of four weighted partial utilities



Figure 4: Factors influencing the agent's adoption-decision process and their representation in the model

Source: Own illustration

and is calculated as follows:

$$U(j) = w_{pp}(sm_j) \cdot u_{pp}(j) + w_{env}(sm_j) \cdot u_{env}(j)$$

$$+ w_{inc}(sm_j) \cdot u_{inc}(j) + w_{com}(sm_j) \cdot u_{com}(j),$$
(1)

where

$$\sum_{k} w_k(sm_j) = 1 \text{ for } k \in K : \{pp, env, inc, com\} \text{ and } w_k(sm_j), U(j) \in [0,1].$$

The partial utilities  $u(\cdot)$  account for the payback period of the investment  $(u_{pp})$ , the environmental benefit of investing in a PV system  $(u_{env})$ , the household's income  $(u_{inc})$ , and the influence of communication with other agents  $(u_{com})$ . Each partial utility is calculated on the basis of specific influence factors (see Figure 4) and is normalized<sup>8</sup> in order to lie within the [0,1] interval. The weights  $w(\cdot)$  assigned to each partial utility vary according to the agent's Sinus-Milieu<sup>®</sup>  $(sm_j)$  and are determined in the model's calibration. Next, we illustrate how each partial utility is calculated.

#### 4.1.1 Economic utility

The estimation of the economic utility of adoption is based on the expected payback period pp of a specific PV system for agent j. The payback period is then converted into a linear utility function whose value ranges between 0 and 1. The utility function is

<sup>&</sup>lt;sup>8</sup>The total utility of an adopter is defined within the [0,1] interval. As a result, all partial utilities need to be normalized. In accordance with Zhao et al. (2011), the utility of the payback period is programmed as a linear function, while all other partial utility functions follow an S-shaped curve, also within the [0,1] interval.

calculated as follows:

$$u_{pp}(j) = \frac{max(pp) - pp(j)}{max(pp) - min(pp)} = \frac{21 - pp(j)}{20}.$$
(2)

In order to ensure that the partial utility arising from the payback period lies within the [0,1] interval, and given that the payback period is calculated over 20 years (i.e. the expected useful life of the PV system), the values corresponding to the minimum (min(pp)) and maximum (max(pp)) payback periods are 1 and 21 years, respectively.

The payback period is determined by the year in which the net present value (NPV) of the PV system turns from negative to positive. The NPV is defined as the sum of the discounted cash flows (R(t)) of the PV system, given the initial investment costs  $(I_0)$  and the interest rate (i):

NPV = 
$$-I_0 + \sum_{t=1}^{20} \frac{R(t)}{(1+i)^t}$$
. (3)

The investment costs are the product of the maximum peak power  $(P_{MMP})$  and the price per installed kW of the PV system  $(p_{PV})$ , such that:

$$I_0 = P_{MMP} \cdot p_{PV}(t_0) \tag{4}$$

$$P_{MMP} = G_{STC} \cdot A_{PV} \cdot \eta_{SC} \cdot \eta_{PV}. \tag{5}$$

The peak power of the PV system is computed by the available rooftop area for PV modules  $(A_{PV})$ , the efficiency of the solar cells  $(\eta_{SC})$ , the PV system efficiency  $(\eta_{PV})$ , and the irradiation at standard conditions  $(G_{STC})$ , which is assumed to equal 1 kW/m<sup>2</sup>. The estimation of the system's NPV at a given time period assumes that the price and efficiency of the PV system remain constant. Note, however, that in each simulation step the price per installed kW of the PV system and the cell's efficiency are exogenously updated (see also section 6.1) . In addition, the available roof area for PV modules depends on the type of housing. All other values are kept constant throughout the simulation.

As shown in eq. (6) below, the cash flow R(t) is composed of five factors. The term  $R_{Save}(t, CE)$  includes all earnings that are generated by directly using the produced electricity instead of buying it from or selling it to the grid provider. The terms  $R_{Gov}(t, CE)$ ,  $R_{Adm}(CE)$ ,  $R_{Main}(t)$ , and  $R_{Deprec}(t)$  indicate cash flows due to governmental support, administrative fees, maintenance and upfront costs, and depreciation allowance payments, respectively.

$$R(t) = R_{Save}(t, CE) + R_{Gov}(t, CE) - R_{Adm}(CE) - R_{Main}(t) - R_{Deprec}(t).$$
(6)

The explicit estimation of the revenues due to electricity savings<sup>9</sup> ( $R_{Save}(t, CE)$ ) is a function of time t and of the governmental policy in place. As a result, the calculation of  $R_{Save}(t, CE)$  varies across the different formulations of the Conto Energia (CE). For the CEs 1-4, the savings are computed by considering the electricity grid as a storage component of the PV system. From the introduction of CE 5 onwards,  $R_{Save}(t, CE)$  is calculated as:

$$R_{Save}(t, CE 5) = E_{PV}(t) \cdot [x_{DC} \cdot p_{elec,buy} \cdot (1 + \tau_{elec,buy})^{t-1} + (1 - x_{DC}) \cdot p_{elec,sell} \cdot (1 + \tau_{elec,sell})^{t-1}].$$

$$(7)$$

The estimated savings are a function of the produced amount of electricity  $(E_{PV}(t))$ , the share of direct electricity consumption  $(x_{DC})$ , and the price of electricity, which varies depending on whether the consumer is selling it to  $(p_{elec,sell})$  or buying it from the grid provider  $(p_{elec,buy})$ .<sup>10</sup> In addition, electricity prices are assumed to grow geometrically at constant rates  $(\tau_{elec,sell} \text{ and } \tau_{elec,buy})$ .<sup>11</sup> The first right-hand side term in eq. (7) describes the cost savings due to direct consumption of the PV-generated electricity. The second term describes the earnings from selling PV electricity to the local energy provider.<sup>12</sup>

Importantly, the amount of electricity  $E_{PV}$  generated by the system is a function of the level of irradiation  $(E_{Sun})$ , of the installed nominal maximum peak power  $(P_{MPP})$ , and of the predicted PV module abrasion<sup>13</sup> ( $\xi_{Abrasion}$ ). Furthermore, the level of irradiation depends on the region where the house is located, such that:

$$E_{PV}(t) = E_{Sun} \cdot P_{MPP} \cdot (1 - \xi_{Abrasion})^{t-1}.$$
(8)

Besides energy savings, an additional positive cash flow is generated by governmental support  $(R_{Gov}(t, CE))$ , which is based on the FiT given by the CE. The amount of the support is calculated as the sum of three components: a basic payment for the production of electricity  $(FiT_{Prod}(CE))$ , an incentive for direct PV electricity consumption  $(FiT_{DC}(CE))$ , and, if applicable, additional bonuses  $(FiT_{Bon}(CE))$  that accrue in special circumstances<sup>14</sup>. The cash flows associated with governmental support are then

<sup>&</sup>lt;sup>9</sup>Electricity may be directly consumed by the owner of the PV system, thus saving part of his/her electricity bill. The owner then sells to the utility provider the surplus PV electricity that is not used for self-consumption.

<sup>&</sup>lt;sup>10</sup>In general, the amount of money paid by local energy providers is only a fraction of the electricity price they charge consumers for electricity consumption.

<sup>&</sup>lt;sup>11</sup>Since there is no price increase for t = 1, the electricity price grows by the power of t - 1.

 $<sup>^{12}</sup>$ Note that the second term is independent from and additional to the governmental feed-in tariffs.

<sup>&</sup>lt;sup>13</sup>Similar to the electricity price, the abrasion increases over time by the power of t - 1.

<sup>&</sup>lt;sup>14</sup>For instance, a bonus is paid if the roof of the house is renewed due to asbestos, if the PV system is located in a village with less than 5,000 inhabitants, or if the PV system consists of components that were produced in Europe. The individual bonuses lead to increments in the FiT that range from 5% to 30%. In the model, bonuses are assumed to increase the basic FiT by about 5% on average.

expressed as follows:

$$R_{Gov}(t, CE) = E_{PV}(t) \cdot \left(FiT_{Prod}(CE) + FiT_{DC}(CE) + FiT_{Bon}(CE)\right).$$
(9)

The adoption of a PV system also entails a series of negative cash flows. Administrative fees  $(R_{Adm}(CE))$  have to be paid to the provider of the electricity grid and depend on the specific CE considered, such that:

$$R_{Adm}(CE) = \begin{cases} 30 \frac{\textcircled{}{e}}{year} & \text{for CE 1-3} \\ 3 \frac{\Huge{}{e}}{kW \cdot year} & \text{for CE 4-5.} \end{cases}$$
(10)

Maintenance and upfront costs  $(R_{Main}(t))$  must also be considered. Upfront costs (e.g., the consultation of a PV expert/adviser) are paid in the first year of the investment, while maintenance costs occur yearly. Both expenditures are estimated to be a fraction of the initial investment costs:

$$R_{Main}(t) = \begin{cases} (\alpha_{upfront} + \alpha_{Main}) \cdot I_0 & \text{if } t = 1\\ \alpha_{Main} \cdot I_0 & \text{otherwise} \end{cases}$$
(11)

Finally, the cash flow includes depreciation allowance payments of the PV system  $(R_{Deprec}(t))$ . The depreciation allowance amounts to a fixed outflow taking place at the end of every year for 20 years, at which point the remaining value of the fixed asset at the end of its useful lifetime is zero.

#### 4.1.2 Environmental utility

The partial utility  $u_{env}(j)$  in eq. (1) is meant to capture an agent's attitude toward the environmental/ecological advantages associated with the adoption of a PV system. These attributes could be measured by the amount of CO<sub>2</sub> emissions saved; however, for reasons of simplicity, the partial utility considers only the expected amount of energy generated by the PV system. In line with Marheineke (2002), we assume that the energy required to produce a PV system is small in comparison to the amount of "green" energy it generates. The actual output of the PV system depends on its location and technical attributes, and the estimated environmental utility is assumed to follow an S-shaped function, where  $E_{PV,tot,j}$  is the expected amount of electricity generated over 20 years by the PV system of agent j, and  $\bar{E}_{PV,tot}$  is the expected average amount of electricity



Figure 5: Utility function of the environmental benefits associated with the adoption of a PV system

Source: Own illustration

generated over 20 years by all PV systems, such that:

$$u_{env}(j) = \frac{\exp\left(\frac{E_{PV,tot,j} - E_{PV,tot}}{1 \cdot 10^4}\right)}{1 + \exp\left(\frac{E_{PV,tot,j} - \bar{E}_{PV,tot}}{1 \cdot 10^4}\right)}.$$
(12)

Figure 5 shows the environmental utility function curve and its operational range. The figure indicates that the environmental utility does not have its minimum at zero. This is due to the fact that PV systems always save energy when operating. However, the curve also implies that the agent becomes less responsive to  $CO_2$  savings as the amount of expected electricity generation increases.

### 4.1.3 Income utility

The partial utility  $u_{inc}(j)$  is based on the household's income, which in turn is determined by the agent's region and his/her socio-demographic attributes. In general, it is assumed that agents with an above-average income are more likely to invest in a PV system. This consideration is accounted for in the functional representation of an agent's income utility, whose S-shaped curve depends on agent j's income  $(N_j)$  and the average income of all agents in the model  $(\bar{N})$ , such that:

$$u_{inc}(j) = \frac{\exp\left(\frac{N_j - \bar{N}}{1 \cdot 10^3}\right)}{1 + \exp\left(\frac{N_j - \bar{N}}{1 \cdot 10^3}\right)}.$$
(13)

#### 4.1.4 Communication utility

Finally, the influence of the social communication network on the adoption decision is represented by the partial utility  $u_{com}(j)$ . In the model it is expressed as a function of agent j's total number of communication links  $(L_{j,tot})$  and in relation to the number of links with actual adopters  $(L_{j,adopter})$ . Since there are no or only a few adopters in the social system at the beginning of the diffusion process, communication hardly plays a role initially. The S-shaped communication utility function therefore starts with a value of about zero and increases as the diffusion process takes place. The partial utility is estimated by the following expression:

$$u_{com}(j) = \frac{\exp\left(\frac{L_{j,adopter} - 0.5 \cdot L_{j,tot}}{0.8}\right)}{1 + \exp\left(\frac{L_{j,adopter} - 0.5 \cdot L_{j,tot}}{0.8}\right)}.$$
(14)

In the model, it is assumed that the total number of links (to both adopters and nonadopters) varies according to the Sinus-Milieu<sup>®</sup> of the agent (see Figure 6). The resulting array of functions represented by eq. (14) nonetheless guarantees that each agent, independently of his/her Sinus-Milieu<sup>®</sup>, has an equal response to a proportional increase in the number of links to other adopters.<sup>15</sup> For a more detailed explanation of the communication network see also section 4.3.

## 4.2 Modeling socio-economic attributes in the PV diffusion process

Investments in a new technology are related not only to economic considerations, but also to specific attitudes towards a technology's attributes (Rogers, 2003). These attitudes are the product, among other things, of an agent's socio-economic background and his/her lifestyle choices.

In the model, the social system is represented by different socio-economic categories. Each category identifies groups of individuals displaying similarities in their socio-economic

<sup>&</sup>lt;sup>15</sup>For instance, consider an agent with six links, three of those links being links to other adopters. The resulting communication utility is 0.5. An agent with ten links, five of which are to adopters, also has a communication utility of 0.5.



Figure 6: Response of the communication utility in relation to different communication network configurations

Source: Own illustration

behavior and consumption patterns. The social system is thereby characterized by subgroups that have common values and attitudes toward work, family, leisure, money, and consumption. Following Schwarz (2007) and Schwarz and Ernst (2009), we incorporate these socio-economic groups and their attitudes toward innovative technologies in the model by referring to Sinus-Milieus<sup>®</sup>.

The Sinus-Milieus<sup>®</sup> include a wide array of social categories that range from the enlightened middle-class ("Borgehsia Illuminata") to consumers-materialists ("Consumisti Precari"). Figure 7 shows the eight Sinus-Milieus<sup>®</sup> modeled in our study and displays them as a function of social status and basic values. The model uses freely available Sinus-Milieu<sup>®</sup> data for Italy from 2003 provided by For Sale Italia Advertising Agency (2004). In addition, the milieus are also associated to the adopter categories defined by Rogers (2003), as illustrated in Table 4.

The Sinus-Milieu<sup>®</sup> structure of the model is created during an initialization phase that proceeds the first simulation period. In the initialization, the agents/households are allocated to the different Italian regions and assigned to various categories. Following the available census statistics (ISTAT, 2012), these categories include six household types, five age classes and five education levels.<sup>16</sup> The model then restricts its focus to agents living in single- or two-family houses (see Figure 8). At this point, four attributes are assigned

<sup>&</sup>lt;sup>16</sup>There is a total of 150 categories spread across the 20 regions considered. Each single category is formally represented in the model by an object in an array, which makes them easily accessible from a programming point of view. The individual agents that are assigned to each category are saved as vectors, which guarantees high computational speed when calculating, for instance, the NPV.



Figure 7: Share of population by Sinus-Milieus<sup>®</sup> in Italy, 2003

Source: Own illustration, based on For Sale Italia Advertising Agency (2004)

to each household: the Sinus-Milieu<sup>®</sup>, the average income, the electricity consumption level, and the type of housing (see Figure 9).

The household income is correlated to the agent's household type, age class, education level, and region. It is assumed that the average household income can be described by a logarithmic probability distribution (Statistisches Bundesamt, 2012), whose standard deviation depends on Italy's Gini coefficient, which is about 0.337 (OECD, 2011)<sup>17</sup>.

The energy consumption of an agent depends on the number of household members, which, in turn, is also associated to the household type. The average number of Italian household members, as well as the average energy consumption per household, is based on statistical data (ISTAT, 2012). In contrast, the average number of household members in each Sinus-Milieu<sup>®</sup> rests on own assumptions.

Finally, the housing type is linked to the household's income. According to Eurostat (2012), the likelihood that an agent lives in a single-family house is significantly higher if his income exceeds the Italian median by 60%. The probability increases even further for

<sup>&</sup>lt;sup>17</sup>The Gini coefficient is a measure of income inequality within a country. It ranges from 0 (perfect equality) to 1 (perfect inequality). The Gini coefficient presented here is based on disposable household income, corrected for household size and deflated by the consumer price index (CPI). Italy displays an intermediate level of income inequality in comparison to other developed countries. The OECD average is 0.314 (OECD, 2011), a value between those of Norway (0.256) or Germany (0.295), and those of the USA (0.378) and Mexico (0.476).

Sinus-Milieu <sup>®</sup>	Adopter categories	Reason for assignment
Borghesia Illuminata	Innovators,	Highest income,
Enlightened Middle Class	Early Adopters	rational-economical thinking
Neo-Achievers	Innovators,	Environmental thinking, high income,
Neo-Achievers	Early Adopters	high knowledge, take risks
Progressisti Tolleranti	Early Adopters,	Intellectuals, basic ecological
Tolerant Progressists	Early Majority	and economic thinking
Italia Media Ambiziosa	Early Majority,	Consider social norms, influenced
Average Middle Class	Late Majority	by mass media communication
Tradizionali Conservatori	Late Majority,	Do not take risks,
Traditional Conservatives	Laggards	adopt only when everyone does
Classe Post Operaria	Early Majority,	Consider social norms, strongly
Working Class	Late Majority	influenced by communication
Edonisti Ribelli	Early Adopters,	See the potential of PV systems
<i>Hedonists</i>	Early Majority	but do not have money
Consumisti Precari	Early Majority,	Strongly influenced
Precarious Consumerists	Late Majority	by peer-to-peer communication

Table 4: Sinus-Milieus<sup>®</sup> and adopter categories in Italy

Source: Own assumptions and illustration, based on For Sale Italia Advertising Agency (2004) and Rogers' (2003) adopter categories

agents living in two-family homes. It is important to differentiate between housing types, as they are associated with different roof areas<sup>18</sup>, which pose a limit to the maximum peak power of the PV system. Since the average household income is higher in Northern Italy, more agents live in single-family houses, resulting in a higher than average PV power per adopter. In Southern Italy there is a higher level of irradiation, but the median income is lower and fewer people live in single-family houses, so that the average PV system is smaller. Importantly, by accounting for two housing types the model has more control over simulation results, thus improving the model's calibration across the different regions.

After its creation in the model's initialization, the social structure is recursively updated at the end of each simulation period. Updating the social structure involves, on the one hand, the model's calibration over the 2006–2010 period and, on the other hand, the implementation of various assumptions about future demographic developments. The latter include forecasts for the Italian population growth and the number of household members. The size of each milieu varies over time in accordance with the changes in the social system. However, the model assumes that the share of each Sinus-Milieu<sup>®</sup> relative

<sup>&</sup>lt;sup>18</sup>A household in a single-family house has, on average, a larger roof area available for PV modules than a household living in a two-family house.



Figure 8: Initialization and structure of the social system in the model

Source: Own illustration

to the total agent populations remains constant from 2010 onwards.<sup>19</sup>

One last remark must be made before the next section is introduced. Once the social system is initialized, the model includes about 10 million Italian households as possible adopters<sup>20</sup>. Since each agent has several attributes and needs to perform a series of operations during the innovation-decision process, the model requires a relatively large computer storage capacity and the simulation time can be long. In order to reduce the computational effort, the model includes the option to scale the number of  $agents^{21}$ , i.e. one agent may represent several households simultaneously, thereby reducing the number of potential adopters and speeding up the simulation process. This process may have implications for the accuracy of the estimations and is discussed in more detail during the model calibration stage described in section 5.

<sup>&</sup>lt;sup>19</sup>While the total number of agents in each milieu may grow, the relative share of each Sinus-Milieu<sup>®</sup> remains constant. This simplification results from a lack of forecasts concerning the future evolution of Sinus-Milieus<sup>®</sup> and is justified by the fact that their share remains almost unvaried during the model's calibration over the 2006–2010 period.

<sup>&</sup>lt;sup>20</sup>In 2006, Italy had a population of about 59.1 million inhabitants and a total of about 23.9 million households, 10 million of which are living in one- or two-family houses.

 $<sup>^{21}</sup>$ Each agent has 20 attribute values, and each attribute value requires about 8 bytes of hard-drive memory. If there are 10 million agents, one simulation step requires about 1.5 GB and the whole simulation needs about 30 GB hard-drive storage capacity. As a result, the simulation lasts longer than 12 hours.



Figure 9: Attributes of a representative agent in the model's social structure

Source: Own illustration

## 4.3 The communication network

The model's social structure also affects the communication among agents, which in turn influences the adoption decision. As done by Schwarz and Ernst (2009), communication channels between agents are assigned according to the Small-World-Network (SWN) algorithm, which was originally created by Watts and Strogatz (1998). SWNs are based on the idea that every individual is connected to anyone else through no more than six degrees of separation (Barabási and Bonabeau, 2003). In addition, SWNs are characterized by a high density of connections with short path-lengths, features also shared with actual social communities. Empirical studies have shown a strong correlation between the number of contacts in a SWN and the agents' gender, age, education, and income (Schwarz, 2007; Zheng et al., 2006).

In the model, the number of communication channels depends on the Sinus-Milieu<sup>®</sup> of the agent. Furthermore, the SWN algorithm has been adjusted in order to account for the structure of the social system considered. All possible adopters are situated across the 20 regions and have primarily "localite" links to other agents from the same region. In addition, most of the communication channels are modeled to take place between agents belonging to the same socio-economic group (see Table 5). The remaining links are almost uniquely with agents from bordering Sinus-Milieus<sup>®</sup> (see Figure 7). Note that the network structure (i.e. the links across specific agents) is created in the initialization of the model and maintained throughout the simulations. However, in order to create an element of uncertainty, there is a small probability in each simulation run that an agent will break up a link and randomly reconnect to another agent (see Table 6).<sup>22</sup>

 $<sup>^{22}</sup>$ For instance, for any of the 6 links to other agents of a Neo-Achiever, there is a 1% chance that the

Sinus-Milieu®	Borghesia Illuminata	Neo-Achievers	Progressisti Tolleranti	Italia Media Ambiziosa	Tradizionali Conservatori	Classe Post Operaria	Edonisti Ribelli	Consumisti Precari
Borghesia Illuminata	85	10	5	0	0	0	0	0
Neo-Achievers	10	75	10	5	0	0	0	0
Progressisti Tolleranti	5	10	70	10	5	0	0	0
Italia Media Ambiziosa	0	5	10	70	10	5	0	0
Tradizionali Conservatori	0	0	5	10	70	10	5	0
Classe Post Operaria	0	0	0	5	10	70	10	5
Edonisti Ribelli	0	0	0	0	5	10	75	10
Consumisti Precari	0	0	0	0	0	5	10	85

**Table 5:** Probabilities to connect to other agents in own and other Sinus-Milieus<sup>®</sup> [%]

Source: Own assumptions, based on Schwarz (2007)

Table 6:	Number of	communication	channels	and	probability	to rando	mly	reconnect
----------	-----------	---------------	----------	-----	-------------	----------	-----	-----------

Sinus-Milieu <sup>®</sup>	Borghesia Illuminata	Neo-Achievers	Progressisti Tolleranti	Italia Media Ambiziosa	Tradizionali Conservatori	Classe Post Operaria	Edonisti Ribelli	Consumisti Precari
Number of links Probability to reconnect [%]	$7\\0.5$	$\begin{array}{c} 6 \\ 1.0 \end{array}$	7 1.0	$\frac{8}{0.75}$	$\begin{array}{c} 6 \\ 0.25 \end{array}$	$\frac{8}{0.5}$	9 1.0	$\begin{array}{c} 10 \\ 0.25 \end{array}$

Source: Own assumptions, based on Schwarz (2007)

# 5 Model calibration

We calibrate the model with respect to the total number of adopters, the rate of adoption, the installed PV power, and the PV system characteristics over the 2006–2011 period. We target a close resemblance of the simulation results with the actual PV diffusion process at the national level. As adjusting the model in order to fit the PV adoption dynamics of

link will be broken and a new connection will be created with another agent.



Figure 10: Outcome of the calibration for different threshold levels, 2006-2011 Source: Own illustration, based on calibration results

each individual region is particularly difficult, each region is individually calibrated once the model matches the general national PV diffusion trends.

The calibration is performed by adjusting the values of the utility threshold and the weights of the partial utilities across the various socio-economic adopter categories. Changes to the partial utility weights of a specific socio-economic group influence the slope of the adopter curves of that given agent's category, thereby affecting their specific attitude towards the innovation. Changes to the utility threshold, in contrast, shape the whole level/slope of the curves without affecting specific adopter categories.

Figure 10 shows the results of the calibration for the total number of adopters and the rate of adoption at the national level. The diagrams illustrate the actual PV market data and various simulation runs with different thresholds, while all other parameters are kept constant. The model displays a good fit to the actual number of adopters. The best results are obtained with a threshold value of 0.539. However, the simulations also turn out to be rather sensitive to variations in the threshold level. A threshold change of  $\pm 0.03$  causes a difference in the number of adopters of about  $\pm 18\%$ , whereas a change of  $\pm 0.06$  leads to fluctuations in the  $\pm 35\%$  range.

The simulated rate of adoption is less accurate in matching the actual PV statistics. This is primarily due to the year 2008. In 2008, investment costs were still relatively high and the introduction of the CE 2, which brought a first reduction in support payments and led to a fall in the NPV of the PV system as well as a longer payback period. In the model, PV systems were not as economically appealing as before, which led to a lower



**Figure 11:** Calibration of the installed PV capacity, 2006–2011 Source: Own illustration, based on calibration results

number of adopters and a lower adoption rate than displayed in the the actual market.<sup>23</sup> Nevertheless, the rate of adoption better resembles the actual values in the following years (i.e. 2009–2011), thus still capturing a key trend to be picked up for the successful prediction of the PV market's future development.

Figure 11 shows the calibration of the total installed PV capacity, the average PV power per adopter, and the average roof surface area of the PV systems. The achieved fit is acceptable for all three parameters. Note that the average roof-surface area of PV systems is assumed to be constant in the model. However, the slightly increasing average installed PV power per adopter is guaranteed thanks to improving PV module efficiency over time.

The partial utility weights implemented in the model are shown in Table 7. They have been determined by trial and error in response to the simulation results during the calibration. Obviously, there may be other value combinations that could help achieve similar or better calibration results. Nevertheless, the chosen values lead to a good fit for most of the Italian regions. Still, it should be explicitly mentioned that the model responds unevenly to changes to different weights. In particular, the weight of the payback period has, due to the linear formulation of its partial utility, a stronger impact on the diffusion process than the other weights. Therefore, the weight coefficients should not be directly compared to each other and their value should be interpreted as their relative importance in the adoption decision process.

In addition, the weights have been assigned so as to replicate the allocation of the

 $<sup>^{23}</sup>$ Note that attempts to overcome this issue by altering the weights of the partial utilities across different adopter categories did not produce significant improvements.

Weights	$W_{pp}$	Wenv	$W_{inc}$	$W_{com}$
Borghesia Illuminata	0.060	0.350	0.300	0.290
Neo-Achievers	0.070	0.350	0.310	0.270
Progressisti Tolleranti	0.150	0.310	0.265	0.275
Italia Media Ambiziosa	0.150	0.310	0.260	0.280
Tradizionali Conservatori	0.140	0.290	0.260	0.310
Classe Post Operaria	0.140	0.310	0.270	0.280
Edonisti Ribelli	0.135	0.310	0.280	0.275
Consumisti Precari	0.125	0.320	0.280	0.275

Table 7: Calibrated weights by Sinus-Milieus<sup>®</sup>

Source: Calibration results



Figure 12: Number of adopters by Sinus-Milieu<sup>®</sup>, 2006–2011

Source: Own illustration, based on calibration results

Sinus-Milieus<sup>®</sup> with respect to Rogers' (2003) adopter categories presented in Table 4. Figure 12 shows the number of new adopters in each Sinus-Milieu<sup>®</sup> between 2006 and 2011. Initially, the diffusion process is driven mainly by innovators and early adopters (2006–2008). Later, as the rate of adoption increases, also the average middle class is participating in the adoption process (2009–2011). As a result, innovators and early adopters are characterized by higher coefficients for the income and environment weight. Small coefficients for the payback period weight indicate that innovators are willing to take more risk. Later adopters are characterized by higher coefficients for the weight of the payback period, thus stressing their need for financial security.

Figure 13 shows the distribution of the Sinus-Milieus<sup>®</sup> over time, regardless of the adoption status. The calibrated distribution of the socio-economic groups fits almost perfectly to the reference values observed in real world data, which is given as a share of households. The milieus are slightly different across the regions and depend on local socio-demographics. The distribution of the Sinus-Milieus<sup>®</sup> changes slightly between



**Figure 13:** Distribution of households according to the Sinus-Milieus<sup>®</sup>, 2006–2011 Source: Own illustration, based on calibration results

2006 and 2011, but no further changes are assumed to take place in the social system (see section 4.2).

Finally, it is important to have a closer look at the option to scale the number of agents implemented in this model and already mentioned in section 4.2. The option works well for rather large regions of Italy with many inhabitants, for example Veneto (see Figure 14a). For these regions, the agent scale may be increased up to 80 without significant effects on the results of the model. In contrast, the fit of the calibration is more problematic for smaller regions with only few inhabitants, e.g. Molise, when the agent scale is large (see Figure 14b). The calibration issue arises as the agent scale approaches or even surpasses the number of agents in one or more categories of the regional social system.

During the calibration, and in the further scenarios of the model, an agent scale of 15 is used. This value keeps the simulation duration and the required computational memory small while limiting the calibration error in small regions to a minimum (see Table 8). As a matter of fact, when focusing on the calibration of the model at the national level, the "agent scale-error" in the small regions has a negligible influence, since the number of adopters is comparably small.

## 6 Scenario analysis and results

After the agent-based diffusion model has been calibrated, it can be used to predict the future Italian PV market under various scenarios. Three simulation scenarios have been tested to consider the sensitivity and validity of the model: a Baseline scenario with the most likely development of the PV market, a scenario with different PV investment costs (Scenario II), and a policy-driven scenario with varying degrees of future governmental



Figure 14: Influence of the agent scale on the number of adopters, 2006–2011

Source: Own illustration, calibration results

Table 8: Influence of the agent scale on the duration<sup>a</sup> of the simulation [s], 2006–2011

Agent scale	5	10	15	20	40	60	80
Veneto <sup>b</sup>	681	342	230	174	91	64	50
Molise <sup>c</sup>	26	18	16	15	13	12	11

<sup>a</sup> Simulations performed with a utility adoption threshold of 0.539

<sup>b</sup> Veneto has about 1.9 to 2.0 million households.

<sup>c</sup> Molise has about 300 to 320 thousand households.

Source: Calibration results

PV support (Scenario III). All three scenarios build on the parametrization obtained from the initial calibration.

## 6.1 Baseline scenario

#### 6.1.1 Description Baseline scenario

The Baseline scenario considers the most likely development of the Italian PV market from 2012 to 2026. Governmental support is modeled on the current CE 5. The Italian government has planned to maintain the CE 5 scheme until the end of 2014. Afterwards, the model assumes that incentives will decrease by 15% every six months. Figure 15 shows the development of the incentive scheme over time.<sup>24</sup>

 $<sup>^{24}</sup>$ Figure 15 shows the average incentive for PV-generated electricity from systems with an installed capacity of up to 20 kW of peak power. Extra payments and payments for direct energy consumption are not included.



Figure 15: Incentive scheme in the Baseline scenario, 2006–2026Source: Own illustration, based on MSE (2005, 2007, 2010, 2011, 2012) and own assumptions

Table 9: Cumulative global installed PV power and PV system price developments, 2012–2026

Year	2012	2013	2014	2015	2020	2026
PV power [GW] System price <sup>a</sup> [ $\in$ /kW]	$77 \\ 1,904$	88 1,824	$100 \\ 1,750$	$125 \\ 1,626$	$345 \\ 1,543$	$760 \\ 1,021$

<sup>a</sup> VAT excluded. Prices refer to small-scale (1–20 kW) PV systems

Source: EPIA (2011) and own calculations

Besides governmental support, investment costs are probably the second most important factor for the future development of the PV market. They play an important role in the estimation of the "Levelized Cost of Electricity" (LCOE) generation, a measure of the value of electricity self-generation. The LCOE of a PV system depends on its investment costs ( $I_0$ ), yearly running costs ( $R_t$ ), financing conditions (i.e. the interest rate *i*), energy output ( $E_{PV}$ ), and economic lifetime (*n*) of the technology (Kost et al., 2012). The LCOE for new PV systems equals the ratio of the total costs of a PV system to the total energy produced over the lifetime of the PV system, measured as:

$$LCOE = \frac{I_0 + \sum_{t=1}^n \frac{R_t}{(1+i)^t}}{\sum_{t=1}^n \frac{E_{PV}}{(1+i)^t}}$$
(15)

Usually, the dominant component of the LCOE of a PV system are its investment costs. About half of the investment costs of a PV system are due to the modules' price, the other half is due to the inverter, cables, monitoring systems, and the installation

costs (Wirth, 2012).<sup>25</sup> The reduction in PV system prices over time can be ascribed to economies of scale as well as learning effects and improvements in efficiency due to research and development activities (Wirth, 2012; EPIA, 2011). Their cost evolution has often been modeled via experience curves (for a literature overview, see van Sark et al., 2010). Here, we also model the evolution of the PV system price  $(I_t)$  at time tby forecasting the price per installed kW power of the system  $(p_{PV}(t))$  with a one-factor experience curve (see also eqs. (3)–(5)). More specifically, it is assumed that  $p_{PV}(t)$  is a function of the global cumulative PV power (ACC(t)), the experience parameter (-b), the price of the system in the base year  $(p_{PV}(t_0))$  and the global cumulative installed capacity in the base year  $(ACC(t_0))$ . The price of the system per installed kW power at the time t is then given by:

$$p_{PV}(t) = p_{PV}(t_0) \cdot \left(\frac{ACC(t)}{ACC(t_0)}\right)^{-b}$$
(16)

$$LR = 1 - 2^{-b} \tag{17}$$

The model implements a learning rate (LR) of 20% until 2020, followed by a reduced rate of 18% until 2026 (EPIA, 2011). Data for the global cumulative installed PV capacity until 2026 are taken from EPIA (2011). The associated PV system price evolution and the cumulative installed capacity are given in Table 6.1.1.

Besides the price per kW, additional assumptions are necessary to estimate changes in the LCOE of a PV system over time. According to Kost et al. (2012), the maintenance cost of a photovoltaic system increases every year by about 2%, with a starting value of circa 1.3% of the initial investment. The intertemporal value of money is discounted at an interest rate of 6% (*i* in eqs. (3) and (15)). In addition, the PV investment is financed by borrowing 70% of the required capital at an interest of 5%. The energy output depends on the region where the PV system is located. Degradation of the PV system is also taken into account and amounts to about 0.3% per year (Kost et al., 2012). In addition, the efficiency of the PV panels is assumed to improve with linear increments by 1.5% per year, which leads to an efficiency increase from 13.5% in 2013 to 16.9% in 2026. Similarly, electricity prices are growing linearly by about 2% per year (Kost et al., 2012).

#### 6.1.2 Results Baseline scenario

The Baseline scenario indicates a stagnation of the diffusion process in all regions. The inflection point of the diffusion process is very distinct and takes place in 2012 (Figure 16). After the rate of adoption reaches its maximum, the number of new adopters decreases quickly from about 280,000 in 2012 to about 6,500 in 2021.

This outcome seems to be consistent with real-world data. According to the latest

<sup>&</sup>lt;sup>25</sup>Also known as Balance of System (BOS) components.



Source: Own illustration, based on simulation results

PV report (GSE, 2013a), the cumulative adoption of PV systems is still growing, but it is slowing down. Between 2007 and 2011, the number of PV installations more than doubled every year. In 2012, for the first time, the total number of new installations was lower than in the previous year. Installed capacity, while still increasing, has also been growing at a slower pace. Between 2010 and 2011, installed capacity grew by 269%, with a marked increase in the average PV system size from 22 to 38.7 kW. Between 2011 and 2012, installed capacity grew by 29%. Similarly, the average size of newly installed PV systems steadily increased between 2007 and 2011<sup>26</sup>, while in 2012 this indicator dropped to values lower than those of 2010. While our simulation results might overestimate the decrease in PV diffusion, the model still seems to capture the recent slow-down of the investments and trend that, also due to the currently unfavorable economic conditions, may be persistent.

The simulation results can be better contextualized with the help of the prediction scenarios proposed by other studies. EPIA (2012), for instance, estimates<sup>27</sup> a cumulative installed PV capacity in Italy of 23,000 MW by 2016. Our model estimates the cumulative installed capacity of small PV systems at about 4,400 MW in 2016, which corresponds to a share of about 19.0% of the total in that year. This number is consistent with the actual share of 15.5% in 2011.<sup>28</sup>

 $<sup>^{26}</sup>$ Note that there was a jump in the average size of newly installed plants in 2011. This change is probably also due to the activation of several large-scale PV plants.

<sup>&</sup>lt;sup>27</sup>We refer to their "moderate scenario". In their "policy-driven" scenario, EPIA (2012) estimate a total installed capacity of 30,800 MW in 2016.

 $<sup>^{28}</sup>$ If we used EPIA's (2012) "policy-driven" scenario, the actual share of domestic installed PV capacity drops to 14.3%.



Figure 17: Average values of key economic indicators in the Baseline scenario, 2006–2026 Source: Own illustration, based on simulation results

A more detailed analysis of the simulated average NPV values of the PV system helps to better explain the results of the model (see Figure 17). At the beginning of 2006, PV systems were not profitable. Thanks to the introduction of government support and decreasing investment costs, the average NPV of photovoltaics grew steadily until 2012, when it reached a value of about  $\leq 15,000.^{29}$  However, starting already in 2011, the incentive scheme has been reduced dramatically. As a consequence, the average NPV decreases to  $\leq 6,534$  by 2019. Moreover, the CE 5 has changed the calculation method for the clearance balance<sup>30</sup> of direct PV electricity consumption. According to TIS Innovation Park (2012), the Italian government made this change on purpose, in order to support direct electricity consumption more strongly. As a result of the support scheme changes, the PV system owner needs an electricity storage component for his PV system to receive the benefit payments. The model, however, does not simulate any such components.<sup>31</sup>

 $<sup>^{29}\</sup>mathrm{We}$  always refer to small PV systems up to 20 kW of peak power.

<sup>&</sup>lt;sup>30</sup>Refers to the savings summand  $(R_{save}(t, CE))$  in the PV systems' cash flow (see eq. 6).

<sup>&</sup>lt;sup>31</sup>The inclusion of a storage capacity would significantly alter the NPV valuation of the system. In addition, it would complicate the decision as to when to consume and when to feed-in the self-generated electricity.



**Figure 18:** Influence of the different weighted partial utilities in the Baseline scenario Source: Simulation results

Interestingly, the rate of adoption and the NPV of the system increase again from 2021 to 2026. The decline in investment costs eventually makes the PV system economically profitable again, despite the small remaining governmental support<sup>32</sup>. The average investment costs, which depend on the given PV system price (Table 6.1.1), are the most important component for the estimation of the production costs of the self-generated electricity. As shown in Figure 17, the model predicts grid parity <sup>33</sup> in 2010. This result corresponds to the actual point in time when residential grid parity was achieved in Italy (Breyer and Gerlach, 2013), thus confirming the good parametrization of the model. After grid parity is reached, the model predicts further reductions in PV electricity production costs, which is in accordance with the assumed decrease in the investment costs and the values forecasted by Breyer and Gerlach (2013).

Understanding the results of the Baseline scenario requires also a closer look at the innovation-decision process of the model. Figure 18 displays the average share of weighted partial utilities of the calibrated model as a function of time. The diagram includes all agents' utilities, regardless of whether they are adopters or not. As one can see, the influence of the communication network is negligible during the calibration phase and increases only marginally from 2010 to 2013. Afterwards, the influence of communication remains constant. The influence of the households' income and the importance of environmental concerns decrease as PV diffusion expands until 2011. In contrast, the share of weight of the payback period increases between 2006-2012, which can be explained by its linear influence on the partial utility, as well as by the strong increase in the NPV in the first years of the simulation.

<sup>&</sup>lt;sup>32</sup>The average payback period follows a curve that is the inverse of the NPV curve, and thus its shape and evolution over time may be explained in a similar way.

 $<sup>^{33}{\</sup>rm Grid}$  parity takes place when electricity from the grid and self-generated PV electricity (i.e., LCOE) have equal production costs.

Overall, the model leads to stable and reproducible simulation results, which nevertheless may be questioned. Especially the influence of each partial utility could have been designed in a different way. However, it seems reasonable that communication networks have a rather small influence on the adoption decision, since the share of adopters to the total agent population remains small across the entire simulation period (max. of 6.6% in 2026). Increasing the weight of the communication utility has little to no influence on the outcome of the model. Communication is therefore not likely to be the driving force behind the diffusion process. Since environmental and income effects do not change much over time, they are also not likely to play a leading role for a potential increment in the diffusion process. The only aspect that may lead to and maintain a high rate of adoption is the economic profitability of the PV system. In order to analyze the model's response to different NPV valuations, the following two scenarios further elaborate on governmental PV support and the price of the PV system.

## 6.2 Scenario II - Changes to the Support Policy

### 6.2.1 Description Scenario II

In this scenario, two alternative governmental incentive schemes are implemented. The Baseline scenario is used as a reference for comparisons. Changes to the support scheme take place from 2015 onwards. While the Baseline scenario considers a decrease in incentives of 15% every six months, here in Scenario II the incentives are reduced by 5% and 25%, respectively. The reduction in the incentive payments leads to an end of governmental support before the last simulation year. The alternative with stronger incentives, instead, guarantees governmental support until 2026 and beyond. Figure 19 displays the alternative support schemes simulated.

### 6.2.2 Results Scenario II

The gradual reductions in the incentive scheme by only 5% increase the number of adopters. Cutting back the incentives by 25% does not contribute to significant differences in the results compared to the Baseline scenario. Both the Baseline scenario and the "weaker incentive" alternative hardly have new adopters between 2015 and 2026. In contrast, the "stronger incentive" program leads to 36% more adopters by the end of 2026 (1,145,900 households) compared to the reference case. Similarly, the cumulative installed PV power increases to 7,900 MW, compared to 4,400 MW in the baseline case.<sup>34</sup> Higher incentives secure a shorter payback period of the investment and incentivizes, from 2015 onwards, at least 31,000 new adopters per year. Nevertheless, the "PV boom" that characterized the 2009–2012 period could not be replicated.

<sup>&</sup>lt;sup>34</sup>The individual regions show similar characteristics as the whole nation and are not further analyzed.



Figure 19: Alternative incentives for the future PV support scheme in Scenario II, 2006–2026 Source: Own assumptions and illustration, based on Conto Energia 5



Figure 20: PV diffusion results in Scenario II, 2006–2026

Source: Own illustration, based on simulation results

The two alternative incentive schemes have a strong impact on the NPV and the payback time of the PV system (see Figure 21). Lower incentives contribute to a drop in the NPV value, which decreases to  $\in 5,240$  in 2018, and then increases again till the end of the simulation period. By 2026, the reduced governmental support scenario shows almost the same NPV as the Baseline scenario. The NPV growth after 2018 is due to decreasing PV investment costs, as it is the case for the baseline simulation. Since in the reference case and in the low incentive alternative the monetary incentives are small and



**Figure 21:** Average values of key economic indicators in Scenario II, 2006–2026 Source: Own illustration, based on simulation results

decreasing, the PV price has a stronger influence on the NPV value<sup>35</sup>. On the contrary, the stronger incentive scheme leads to an almost linearly increasing NPV from 2015 till the end of the simulation period. The final NPV in 2026 is about  $\in 11,890$ .

Besides the two alternative incentive schemes presented here, other governmental support programs have been tested to explore the "boundary behavior" of the model. Cutting off the incentives totally in January 2013 leads to a result similar to the one obtained with weaker incentives. Maintaining the governmental payments of December 2012 throughout the remaining simulation runs also leads to a similar turning point in the rate of adoption as the one obtained with the "higher incentive" scheme, though the NPV in 2026 is higher. In general, the simulations show that the adoption behavior of the agents can be strongly influenced by the incentive scheme adopted by the government. Small to no incentives lead to a stagnation of the diffusion process. Strong incentive programs, in contrast, rapidly accelerate the diffusion dynamics.

## 6.3 Scenario III - Changes to the Investment Costs

### 6.3.1 Description Scenario III

The third scenario simulates two alternative PV system price developments. Both alternatives are derived from the experience curve model adopted for the PV system price forecast (see eqs. (16)-(17)). The learning rates are kept constant for different estimates

 $<sup>^{35}</sup>$ As shown in eq. (3), investment costs in the NPV calculation correspond to a single down payment, while the cash flows (including the support incentives) are discounted over time. As a result, the investment costs have a much stronger direct influence on the final value of the NPV estimation.

Year		2012	2013	2014	2015	2020	2026
Low PV price Change relative to baseline	[€/kW] [%]	$1,904 \\ 0$	1,736 -4.8	1,615 -7.7	1,524 -6.2	967 -17.6	842 -17.6
High PV price Change relative to baseline	[€/kW] [%]	$1,904 \\ 0$	$1,833 \\ +0.5$	1,784 + 1.9	$1,749 \\ +7.6$	$1,543 \\ +31.4$	$1,342 \\ +31.4$

 Table 10:
 Forecasted investment costs in Scenario III, 2012–2026

Source: EPIA (2011)



Figure 22: PV diffusion results in Scenario III, 2006–2026

Source: Own illustration, based on simulation results

of cumulative global installed PV capacity. For the prediction of the cumulative installed PV capacity, EPIA (2011) provides two additional scenarios based on an optimistic or a pessimistic outlook regarding future PV market development. Table 10 lists these investment cost projections as "low" and "high" PV system price alternatives. Moreover, the table shows the percentage change in relation to the original baseline investment costs.

### 6.3.2 Results Scenario III

The results indicate clear differences relative to the Baseline scenario. An incremental reduction in the investment costs of up to 17.6% by 2026 leads to an increase in the total number of adopters by 26.1% to about 1,062,540 households. The share of adopters corresponds to about 8.3% of the total agent population. The higher number of adopters also raises the total installed PV capacity to 7,300 MW. In contrast, the pessimistic scenario regarding investment costs stops the diffusion process. The rate of adoption becomes almost zero and the number of total adopters remains constant from 2013 onwards.



Figure 23: Average values of key economic indicators in Scenario III, 2006–2026 Source: Own simulation, based on simulation results

Compared to the Baseline scenario, this alternative has 11.7% less adopters in 2026 and a total PV power of only 5,100 MW.<sup>36</sup>

The simulation outcome may be explained by looking at the relevant economic parameters that drive the diffusion process of the PV system (Figure 23). Higher investment costs bring about a decrease in the NPV by 29.3% until 2026. A decrease in the PV price by 17.6%, in contrast, increases the NPV by 16.0% at the end of the simulation. The payback period of the PV system and the cost of self-produced electricity follow similar paths.

By comparing the results of Scenario II and Scenario III, it may be argued that both governmental incentives and the evolution of the PV system price have a significant influence on the adoption process. Based on the simulations' outcome, the scenario with the highest incentive scheme obtained the largest technology adoption. Obviously, a oneto-one comparison of the two scenarios is hindered by the many assumptions made. In particular, the PV system price is assumed to depend on the success of PV adoption at a global scale, not forgetting the imputed economies of scale and learning effects

<sup>&</sup>lt;sup>36</sup>For both alternatives, the individual regions show similar characteristics and are not further analyzed.

of the experience curve model. In contrast, incentives can be used more flexibly, as they are directly determined by government policy. As a result, though more expensive to taxpayers, they are a better controllable option to accelerate the diffusion of PV technology.

## 7 Conclusion

While the expansion of large PV sytems may continue, Italy's domestic<sup>37</sup> PV installations have already surpassed an initial phase of rapid growth and, although likely to spread further, they are expected to do so at a significantly slower rate. According to the simulation results, the number of new households adopting photovoltaic technology stagnates under the current support scheme.

In an attempt to adequately account for the complexity of the actual diffusion process of the PV technology, we implement an agent-based model that incorporates four elements influencing the adoption decision: the economic profitability of the investment, environmental considerations, a household's income, and the impact of communication networks. To do so, the model structures the social system into socio-economic classes (Sinus-Milieus<sup>®</sup>). In total, 150 categories across 20 regions have been implemented by distinguishing between age classes, the level of education, and the household type.

Despite the multiple factors interacting simultaneously, the model simulates reproducible and reasonable results that are in line with observed data over the 2006–2011 period. Overall, the calibration of the model proved to be relatively easy to handle by varying the weights of the innovation-decision process and the utility threshold. The projected diffusion can therefore be evaluated by altering key parameters driving the outcome of the model.

As one might expect, it has been shown that the economic profitability of the investment is the most influential criterion in the adoption decision. As a result, we examined in greater detail the two parameters that most influence it: alternative governmental support schemes and variations in the PV system's investment costs. Compared to the Baseline scenario, a steeper reduction in the support payments would stop the diffusion process at once. On the contrary, a more gentle step-wise decline in the incentive scheme would ensure a greater diffusion of the PV technology. Nevetherless, in the simulation results, the adoption rate that characterized the initial diffusion is never replicated under the latest support policy scheme. A similar outcome is obtained with variations in the expected evolution of the PV investment costs. In general, it may be argued that direct governmental support is more costly to taxpayers, but it is a relatively safe option to ensure a speedier diffusion of photovoltaic technology among private homeowners.

 $<sup>^{37}</sup>$ In the model, the agent population contains only households living in single- or two-family houses; hence, only the diffusion of small residential PV systems of up to 20 kW power is considered.

Interestingly, the model managed to accurately predict when grid parity is reached in the Italian market. The model also indicates that self-produced electricity becomes increasingly more advantageous over time. However, Italy witnessed a boom in PV adoption under the influence of strong governmental incentives. Relatively high NPV values were associated with the fast diffusion of the technology. As a result, despite the decline in investment costs and the increasing benefits associated with PV electricity selfgeneration and direct consumption, the agents do not manage to replicate the profitability levels witnessed during the initial PV boom due the significant reduction in support granted by the government. Under the assumption that the preferences of investors will not significantly change over time, the lower profitability of PV systems ultimately explains the reduction in new adoption following the introduction of the new support scheme. Environmental concerns and communication also play an important role, but they are not nearly as significant as economic considerations.

Obviously, the model is built on a number of simplifications and assumptions that fundamentally put into question the validity of its predictions. In particular, as already mentioned, the model is suspiciously sensible to changes in the utility threshold parameter. Small changes in values contribute to strong changes in the diffusion process. In addition, the categorization according to Sinus-Milieus<sup>®</sup> is an effective way to represent the multi-faceted aspects of the current social structure. However, its parametrization in the model was rather *ad-hoc* and not substantiated by verifiable empirical research. While keeping these points in mind, which are shared by many forecasting frameworks, the model's ability to match the actual diffusion of PV systems in Italy at both the national and regional level are encouraging signs of its potential. Furthermore, the applicability of the proposed framework to other countries and, with small changes, to other renewable energy technologies, calls for future implementations with an improved set of underlying parameters.

## References

- Ayompe, L., Duffy, A., McCormack, S., and Conlon, M. (2010). Projected costs of a gridconnected domestic PV system under different scenarios in Ireland, using measured data from a trial installation. *Energy Policy*, 38(7):3731–3743.
- Barabási, A. and Bonabeau, E. (2003). Scale-free networks. *Scientific American*, 255(5):60–69.
- Bass, F. (1969). A new product growth for model consumer durables. *Management Science*, 15(5):215–227.
- Bonabeau, E. (2002). Agent-based modeling: Methods and techniques for simulating

human systems. Proceedings of the National Academy of Sciences of the United States of America, 99(Suppl 3):7280–7287.

- Breyer, C. and Gerlach, A. (2013). Global overview on grid-parity. *Progress in Photo*voltaics: Research and Applications, 21(1):121–136.
- Cantono, S. and Silverberg, G. (2009). A percolation model of eco-innovation diffusion: The relationship between diffusion, learning economies and subsidies. *Technological Forecasting and Social Change*, 76(4):487–496.
- Chow, G. (1967). Technological change and demand for consumers. *American Economic Review*, 57(5):1117–1130.
- Dawid, H. (2006). Agent-based models of innovation and technological change. In Tesfatsion, L. and Judd, K., editors, *Handbook of Computational Economics*, chapter 25, pages 1235–1272. North-Holland, Amsterdam, The Netherlands.
- EPIA (2011). Solar generation 6 Solar photovoltaic electricity empowering the world. European Photovoltaic Industry Association (EPIA), Brussels, Belgium. Available at http://www.epia.org.
- EPIA (2012). Global market outlook for photovoltaics until 2016. European Photovoltaic Industry Association (EPIA), Brussels, Belgium. Available at http://www.epia.org.
- Eurostat (2012). Distribution of population by degree of urbanisation, dwelling type and income group. Available at http://epp.eurostat.ec.europa.eu/.
- Faber, A., Valente, M., and Janssen, P. (2010). Exploring domestic micro-cogeneration in the Netherlands: An agent-based demand model for technology diffusion. *Energy Policy*, 38(6):2736–2775.
- Faiers, A. and Neame, C. (2005). Consumer attitudes towards domestic solar power systems. *Energy Policy*, 34(14):1797–1806.
- Faiers, A., Neame, C., and Cook, M. (2007). The adoption of domestic solar-power systems: Do consumers assess product attributes in a stepwise process? *Energy Policy*, 35(6):3418–3423.
- For Sale Italia Advertising Agency (2004). Gli Italiani. Menschen in Italien. For Sale Lifestyle-Studie für Italien. Caldaro/Kaltern, South Tyrol, Italy.
- Fourt, L. and Woodlock, J. (1960). Early prediction of market success for new grocery products. *Journal of Marketing*, 25(2):31–38.

- Gallo, C. and De Bonis, M. (2013). A neural network model for forecasting photovoltaic deployment in Italy. International Journal of Sustainable Energy and Environment, 1(1):1–13.
- GSE (2012a). Rapporto statistico 2011 Impianti a fonti rinnovabili. Gestore Servizi Energetici (GSE), Rome, Italy. Available at www.gse.it.
- GSE (2012b). Rapporto statistico 2011 Solare fotovoltaico. Gestore Servizi Energetici (GSE), Rome, Italy. Available at www.gse.it.
- GSE (2012c). Totale dei risultati del Conto Energia (Primo, Secondo, Terzo, Quarto e Quinto Conto Energia) - Ripartizione per regione e classe di potenza degli impianti in esercizio. Aggiornamento al 31 Ottobre 2012. Gestore Servizi Energetici (GSE), Rome, Italy. Available at www.gse.it.
- GSE (2013a). Rapporto statistico 2012 Solare fotovoltaico. Gestore Servizi Energetici (GSE), Rome, Italy. Available at www.gse.it.
- GSE (2013b). Totale dei risultati del Conto Energia (Primo, Secondo, Terzo, Quarto e Quinto Conto Energia) - Ripartizione per regione e classe di potenza degli impianti in esercizio. Aggiornamento al 30 Aprile 2013. Gestore Servizi Energetici (GSE), Rome, Italy. Available at www.gse.it.
- Guidolin, M. and Mortarino, C. (2010). Cross-country diffusion of photovoltaic systems: Modelling choices and forecasts for national adoption patterns. *Technological Forecast*ing and Social Change, 77(2):279–296.
- ISTAT (2012). Istituto Nazionale di Statistica (ISTAT), Rome, Italy. Available at http: //www.istat.it.
- Jager, W. (2006). Stimulating the diffusion of photovoltaic systems: A behavioural perspective. *Energy Policy*, 34(14):1935–1943.
- Kiesling, E., Günther, M., Stummer, C., and Wakolbinger, L. (2012). Agent-based simulation of innovation diffusion: A review. *Central European Journal of Operations Research*, 20(2):183–230.
- Kost, C., Schlegl, T., Thomsen, J., Nold, S., and Mayer, J. (2012). Studie Stromgestehungskosten Erneuerbare Energien. Fraunhofer-Institut f
  ür Solare Energiesysteme ISE, Freiburg, Germany.
- Mahajan, V., Muller, E., and Wind, Y., editors (2000). *New-Product Diffusion Models*. Springer-Verlag, Berlin, Germany.

- Mansfield, E. (1961). Technical change and the rate of imitation. *Econometrica*, 29(4):741–766.
- Marheineke, T. (2002). Lebenszyklusanalyse fossiler, nuklearer und regenerativer Stromerzeugungstechniken. Institut für Energiewirtschaft und Rationelle Energieanwendung (IER), Forschungsbericht, Band 87, Universität Stuttgart, Stuttgart, Germany.
- Meade, N. and Islam, T. (2006). Modelling and forecasting the diffusion of innovation a 25-year review. *International Journal of Forecasting*, 22(3):519–545.
- MSE (2005). Primo conto energia, Ministero dello Sviluppo Economico (MSE). Decreto del 14 novembre 2005, Gazzetta Ufficiale No. 181 del 19 novembre 2005. Italian Ministry of Economic Development, Rome, Italy.
- MSE (2007). Secondo conto energia, Ministero dello Sviluppo Economico (MSE). Decreto del 19 febbraio 2007, Gazzetta Ufficiale No. 45 del 23 febbraio 2007. Italian Ministry of Economic Development, Rome, Italy.
- MSE (2010). Terzo conto energia, Ministero dello Sviluppo Economico (MSE). Decreto del 6 agosto 2010, Gazzetta Ufficiale No. 197 del 24 agosto 2010. Italian Ministry of Economic Development, Rome, Italy.
- MSE (2011). Quarto conto energia, Ministero dello Sviluppo Economico (MSE). Decreto del 15 aprile 2011, Gazzetta Ufficiale No. 109 del 15 maggio 15 2011. Italian Ministry of Economic Development, Rome, Italy.
- MSE (2012). Quinto Conto Energia, Ministero dello Sviluppo Economico (MSE). Decreto del 5 luglio 2012, Gazzetta Ufficiale No. 159 del 10 luglio 2012. Italian Ministry of Economic Development, Rome, Italy.
- OECD (2011). An overview of growing income inequalities in OECD countries: Main findings. Organisation for Economic Co-operation and Development (OECD), Paris, France. Available at http://www.oecd.org/els/soc/49499779.pdf.
- Rogers, E. (1962). *The Diffusion of Innovations*. The Free Press, New York, USA, 1st edition.
- Rogers, E. (2003). *The Diffusion of Innovations*. The Free Press, New York, USA, 5th edition.
- Schwarz, N. (2007). Umweltinnovationen und Lebensstile Eine raumbezogene, empirisch fundierte Multi-Agenten-Simulation. PhD thesis, Universität Kassel. Published by Metropolis-Verlag, Marburg, Germany.

- Schwarz, N. and Ernst, A. (2009). Agent-based modeling of the diffusion of environmental innovations - An empirical approach. *Technological Forecasting and Social Change*, 76(4):497–511.
- Schwoon, M. (2006). Simulating the adoption of fuel cell vehicles. *Journal of Evolutionary Economics*, 16(4):435–472.
- Sinus-Institut (2011). Die Sinus-Milieus<sup>®</sup>. SINUS Markt- und Sozialforschung GmbH, Heidelberg, Germany. Available at http://www.sinus-institut.de.
- Sorda, G., Sunak, Y., and Madlener, R. (2013). An agent-based spatial simulation to evaluate the promotion of electricity from agricultural biogas plants in Germany. *Ecological Economics*, 89(5):43–60.
- Statistisches Bundesamt (2012). Definition für die EVS und Leben in Europa (EU-SILC). Available at http://www.destatis.de.
- TIS Innovation Park (2012). Informationsrundschreiben: Der fünfte "Conto Energia". Available at http://www.tis.bz.it.
- van Sark, W., Nemet, G., Schaeffer, G., and Alsema, E. (2010). Photovoltaic solar energy. In Junginger, M., van Sark, W., and Faaij, A., editors, *Technological Learning in the Energy Sector. Lessons for Policy, Industry and Science*, chapter 8, pages 93–114. Edward Elgar, Cheltenham, UK.
- Watts, D. and Strogatz, S. (1998). Collective dynamics of 'small-world' networks. *Nature*, 393(6684):409–410.
- Wirth, H. (2012). Aktuelle Fakten zur Photovoltaik in Deutschland. Fraunhofer-Institut für Solare Energiesysteme ISE, Freiburg, Germany.
- Yang, C. (2010). Reconsidering solar grid parity. *Energy Policy*, 38(7):3270–3273.
- Yuan, X., Zuo, J., and Ma, C. (2011). Social acceptance of solar energy technologies in China - End users' perspective. *Energy Policy*, 39(3):1031–1036.
- Zhai, P. and Williams, E. (2012). Analyzing consumer acceptance of photovoltaics (PV) using fuzzy logic model. *Renewable Energy*, 41(1):350–357.
- Zhang, T., Gensler, S., and Garcia, R. (2011). A study of the diffusion of alternative fuel vehicles: An agent-based modeling approach. *Journal of Product Innovation Management*, 28(2):152–168.
- Zhang, T. and Nuttall, W. (2011). Evaluating the government's policies on promoting smart metering diffusion in retail electricity markets via agent-based simulation. *Journal of Product Innovation Management*, 28(2):169–186.

- Zhao, J., Mazhari, E., Celik, N., and Son, Y. (2011). Hybrid agent-based simulation for policy evaluation of solar power generation systems. *Simulation Modelling Practice* and Theory, 19(10):2189–2205.
- Zheng, T., Salganik, M., and Gelman, A. (2006). How many people do you know in prison? Using overdispersion in count data to estimate the structure in networks. *Journal of the American Statistical Association*, 101(474):409–423.







# List of FCN Working Papers

### 2013

- Grieser B., Madlener R., Sunak Y. (2013). Economics of Small Wind Power Plants in Urban Settings: An Empirical Investigation for Germany, FCN Working Paper No. 1/2013, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, January.
- Madlener R., Specht J.M. (2013). An Exploratory Economic Analysis of Underground Pumped-Storage Hydro Power Plants in Abandoned Coal Mines, FCN Working Paper No. 2/2013, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, February.
- Kroniger D., Madlener R. (2013). Hydrogen Storage for Wind Parks: A Real Options Evaluation for an Optimal Investment in More Flexibility, FCN Working Paper No. 3/2013, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, February.
- Petersen C., Madlener R. (2013). The Impact of Distributed Generation from Renewables on the Valuation and Marketing of Coal-Fired and IGCC Power Plants, FCN Working Paper No. 4/2013, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, February.
- Oberst C.A., Oelgemöller J. (2013). Economic Growth and Regional Labor Market Development in German Regions: Okun's Law in a Spatial Context, FCN Working Paper No. 5/2013, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, March.
- Harmsen van Hout M.J.W., Ghosh G.S., Madlener R. (2013). An Evaluation of Attribute Anchoring Bias in a Choice Experimental Setting. FCN Working Paper No. 6/2013, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, April.
- Harmsen van Hout M.J.W., Ghosh G.S., Madlener R. (2013). The Impact of Green Framing on Consumers' Valuations of Energy-Saving Measures. FCN Working Paper No. 7/2013, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, April.
- Rosen C., Madlener R. (2013). An Experimental Analysis of Single vs. Multiple Bids in Auctions of Divisible Goods, FCN Working Paper No. 8/2013, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, April.
- Palmer J., Sorda G., Madlener R. (2013). Modeling the Diffusion of Residential Photovoltaic Systems in Italy: An Agent-based Simulation, FCN Working Paper No. 9/2013, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, May.

- Ghosh G., Shortle J. (2012). Managing Pollution Risk through Emissions Trading, FCN Working Paper No. 1/2012, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, January.
- Palzer A., Westner G., Madlener M. (2012). Evaluation of Different Hedging Strategies for Commodity Price Risks of Industrial Cogeneration Plants, FCN Working Paper No. 2/2012, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, March (revised March 2013).
- Sunak Y., Madlener R. (2012). The Impact of Wind Farms on Property Values: A Geographically Weighted Hedonic Pricing Model, FCN Working Paper No. 3/2012, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, May (revised March 2013).
- Achtnicht M., Madlener R. (2012). Factors Influencing German House Owners' Preferences on Energy Retrofits, FCN Working Paper No. 4/2012, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, June.

- Schabram J., Madlener R. (2012). The German Market Premium for Renewable Electricity: Profitability and Risk of Self-Marketing, FCN Working Paper No. 5/2012, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, July.
- Garbuzova M., Madlener R. (2012). Russia's Emerging ESCO Market: Prospects and Barriers for Energy Efficiency Investments, FCN Working Paper No. 6/2012, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, July (revised December 2012).
- Rosen C., Madlener R. (2012). Auction Design for Local Reserve Energy Markets, FCN Working Paper No. 7/2012, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, July (revised March 2013).
- Sorda G., Madlener R. (2012). Cost-Effectiveness of Lignocellulose Biorefineries and their Impact on the Deciduous Wood Markets in Germany. FCN Working Paper No. 8/2012, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, September.
- Madlener R., Ortlieb C. (2012). An Investigation of the Economic Viability of Wave Energy Technology: The Case of the Ocean Harvester, FCN Working Paper No. 9/2012, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, October.
- Hampe J., Madlener R. (2012). Economics of High-Temperature Nuclear Reactors for Industrial Cogeneration, FCN Working Paper No. 10/2012, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, October.
- Knaut A., Madlener R., Rosen C., Vogt C. (2012). Impact of Temperature Uncertainty on the Economic Valuation of Geothermal Projects: A Real Options Approach, FCN Working Paper No. 11/2012, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, November.
- Hünteler J., Niebuhr C.F., Schmidt T.S., Madlener R., Hoffmann V.H. (2012). Financing Feed-in Tariffs in Developing Countries under a Post-Kyoto Climate Policy Regime: A Case Study of Thailand, FCN Working Paper No. 12/2012, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, December.
- Blass N., Madlener R. (2012). Structural Inefficiencies and Benchmarking of Water Supply Companies in Germany, FCN Working Paper No. 13/2012, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, December.
- Madlener R., Schabram J. (2012). Predicting Reserve Energy from New Renewables by Means of Principal Component Analysis and Copula Functions, FCN Working Paper No. 14/2012, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, December.
- Harzendorf F., Madlener R. (2012). Optimal Investment in Gas-Fired Engine-CHP Plants in Germany: A Real Options Approach, FCN Working Paper No. 15/2012, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, December.
- Schmitz M., Madlener R. (2012). Economic Feasibility of Kite-Based Wind Energy Powerships with CAES or Hydrogen Storage, FCN Working Paper No. 16/2012, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, December.
- Dergiades T., Christofidou G., Madlener R. (2012). The Nexus between Natural Gas Spot and Futures Prices at NYMEX: What about Non-Linear Causality?, FCN Working Paper No. 17/2012, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, December.
- Rohlfs W., Madlener R. (2012). Assessment of Clean-Coal Strategies: The Questionable Merits of Carbon Capture-Readiness, FCN Working Paper No. 18/2012, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, December.
- Wüstemeyer C., Bunn D., Madlener R. (2012). Bridging the Gap between Onshore and Offshore Innovations by the European Wind Power Supply Industry: A Survey-based Analysis, FCN Working Paper No. 19/2012, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, December.
- Fuhrmann J., Madlener R. (2012). Evaluation of Synergies in the Context of European Multi-Business Utilities, FCN Working Paper No. 20/2012, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, December.

- Sorda G., Sunak Y., Madlener R. (2011). A Spatial MAS Simulation to Evaluate the Promotion of Electricity from Agricultural Biogas Plants in Germany, FCN Working Paper No. 1/2011, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, January (revised October 2012).
- Madlener R., Hauertmann M. (2011). Rebound Effects in German Residential Heating: Do Ownership and Income Matter?, FCN Working Paper No. 2/2011, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, February.
- Garbuzova M., Madlener R. (2011). Towards an Efficient and Low-Carbon Economy Post-2012: Opportunities and Barriers for Foreign Companies in the Russian Market, FCN Working Paper No. 3/2011, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, February (revised July 2011).
- Westner G., Madlener R. (2011). The Impact of Modified EU ETS Allocation Principles on the Economics of CHP-Based District Heating Networks. FCN Working Paper No. 4/2011, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, February.
- Madlener R., Ruschhaupt J. (2011). Modeling the Influence of Network Externalities and Quality on Market Shares of Plug-in Hybrid Vehicles, FCN Working Paper No. 5/2011, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, March.
- Juckenack S., Madlener R. (2011). Optimal Time to Start Serial Production: The Case of the Direct Drive Wind Turbine of Siemens Wind Power A/S, FCN Working Paper No. 6/2011, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, March.
- Madlener R., Sicking S. (2011). Assessing the Economic Potential of Microdrilling in Geothermal Exploration, FCN Working Paper No. 7/2011, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, April.
- Bernstein R., Madlener R. (2011). Responsiveness of Residential Electricity Demand in OECD Countries: A Panel Cointegration and Causality Analysis, FCN Working Paper No. 8/2011, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, April.
- Michelsen C.C., Madlener R. (2011). Homeowners' Preferences for Adopting Residential Heating Systems: A Discrete Choice Analysis for Germany, FCN Working Paper No. 9/2011, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, May (revised January 2012).
- Madlener R., Glensk B., Weber V. (2011). Fuzzy Portfolio Optimization of Onshore Wind Power Plants. FCN Working Paper No. 10/2011, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, May.
- Glensk B., Madlener R. (2011). Portfolio Selection Methods and their Empirical Applicability to Real Assets in Energy Markets. FCN Working Paper No. 11/2011, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, May.
- Kraas B., Schroedter-Homscheidt M., Pulvermüller B., Madlener R. (2011). Economic Assessment of a Concentrating Solar Power Forecasting System for Participation in the Spanish Electricity Market, FCN Working Paper No. 12/2011, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, May.
- Stocker A., Großmann A., Madlener R., Wolter M.I., (2011). Sustainable Energy Development in Austria Until 2020: Insights from Applying the Integrated Model "e3.at", FCN Working Paper No. 13/2011, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, July.
- Kumbaroğlu G., Madlener R. (2011). Evaluation of Economically Optimal Retrofit Investment Options for Energy Savings in Buildings. FCN Working Paper No. 14/2011, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, September.
- Bernstein R., Madlener R. (2011). Residential Natural Gas Demand Elasticities in OECD Countries: An ARDL Bounds Testing Approach, FCN Working Paper No. 15/2011, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, October.
- Glensk B., Madlener R. (2011). Dynamic Portfolio Selection Methods for Power Generation Assets, FCN Working Paper No. 16/2011, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, November.

- Michelsen C.C., Madlener R. (2011). Homeowners' Motivation to Adopt a Residential Heating System: A Principal Component Analysis, FCN Working Paper No. 17/2011, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, November (revised January 2013).
- Razlaf J., Madlener R. (2011). Performance Measurement of CCS Power Plants Using the Capital Asset Pricing Model, FCN Working Paper No. 18/2011, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, November.
- Himpler S., Madlener R. (2011). Repowering of Wind Turbines: Economics and Optimal Timing, FCN Working Paper No. 19/2011, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, November.
- Hackbarth A., Madlener R. (2011). Consumer Preferences for Alternative Fuel Vehicles: A Discrete Choice Analysis, FCN Working Paper No. 20/2011, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, December (revised December 2012).
- Heuser B., Madlener R. (2011). Geothermal Heat and Power Generation with Binary Plants: A Two-Factor Real Options Analysis, FCN Working Paper No. 21/2011, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, December.
- Rohlfs W., Madlener R. (2011). Multi-Commodity Real Options Analysis of Power Plant Investments: Discounting Endogenous Risk Structures, FCN Working Paper No. 22/2011, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, December (revised July 2012).

- Lang J., Madlener R. (2010). Relevance of Risk Capital and Margining for the Valuation of Power Plants: Cash Requirements for Credit Risk Mitigation, FCN Working Paper No. 1/2010, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, February.
- Michelsen C.C., Madlener R. (2010). Integrated Theoretical Framework for a Homeowner's Decision in Favor of an Innovative Residential Heating System, FCN Working Paper No. 2/2010, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, February.
- Harmsen van Hout M.J.W., Herings P.J.-J., Dellaert B.G.C. (2010). The Structure of Online Consumer Communication Networks, FCN Working Paper No. 3/2010, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, March.
- Madlener R., Neustadt I. (2010). Renewable Energy Policy in the Presence of Innovation: Does Government Pre-Commitment Matter?, FCN Working Paper No. 4/2010, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, April (revised June 2010 and December 2011).
- Harmsen van Hout M.J.W., Dellaert B.G.C., Herings, P.J.-J. (2010). Behavioral Effects in Individual Decisions of Network Formation: Complexity Reduces Payoff Orientation and Social Preferences, FCN Working Paper No. 5/2010, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, May.
- Lohwasser R., Madlener R. (2010). Relating R&D and Investment Policies to CCS Market Diffusion Through Two-Factor Learning, FCN Working Paper No. 6/2010, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, June.
- Rohlfs W., Madlener R. (2010). Valuation of CCS-Ready Coal-Fired Power Plants: A Multi-Dimensional Real Options Approach, FCN Working Paper No. 7/2010, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, July.
- Rohlfs W., Madlener R. (2010). Cost Effectiveness of Carbon Capture-Ready Coal Power Plants with Delayed Retrofit, FCN Working Paper No. 8/2010, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, August (revised December 2010).
- Gampert M., Madlener R. (2010). Pan-European Management of Electricity Portfolios: Risks and Opportunities of Contract Bundling, FCN Working Paper No. 9/2010, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, August.
- Glensk B., Madlener R. (2010). Fuzzy Portfolio Optimization for Power Generation Assets, FCN Working Paper No. 10/2010, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, August.
- Lang J., Madlener R. (2010). Portfolio Optimization for Power Plants: The Impact of Credit Risk Mitigation and Margining, FCN Working Paper No. 11/2010, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, September.

- Westner G., Madlener R. (2010). Investment in New Power Generation Under Uncertainty: Benefits of CHP vs. Condensing Plants in a Copula-Based Analysis, FCN Working Paper No. 12/2010, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, September.
- Bellmann E., Lang J., Madlener R. (2010). Cost Evaluation of Credit Risk Securitization in the Electricity Industry: Credit Default Acceptance vs. Margining Costs, FCN Working Paper No. 13/2010, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, September (revised May 2011).
- Ernst C.-S., Lunz B., Hackbarth A., Madlener R., Sauer D.-U., Eckstein L. (2010). Optimal Battery Size for Serial Plug-in Hybrid Vehicles: A Model-Based Economic Analysis for Germany, FCN Working Paper No. 14/2010, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, October (revised June 2011).
- Harmsen van Hout M.J.W., Herings P.J.-J., Dellaert B.G.C. (2010). Communication Network Formation with Link Specificity and Value Transferability, FCN Working Paper No. 15/2010, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, November.
- Paulun T., Feess E., Madlener R. (2010). Why Higher Price Sensitivity of Consumers May Increase Average Prices: An Analysis of the European Electricity Market, FCN Working Paper No. 16/2010, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, November.
- Madlener R., Glensk B. (2010). Portfolio Impact of New Power Generation Investments of E.ON in Germany, Sweden and the UK, FCN Working Paper No. 17/2010, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, November.
- Ghosh G., Kwasnica A., Shortle J. (2010). A Laboratory Experiment to Compare Two Market Institutions for Emissions Trading, FCN Working Paper No. 18/2010, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, November.
- Bernstein R., Madlener R. (2010). Short- and Long-Run Electricity Demand Elasticities at the Subsectoral Level: A Cointegration Analysis for German Manufacturing Industries, FCN Working Paper No. 19/2010, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, November.
- Mazur C., Madlener R. (2010). Impact of Plug-in Hybrid Electric Vehicles and Charging Regimes on Power Generation Costs and Emissions in Germany, FCN Working Paper No. 20/2010, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, November.
- Madlener R., Stoverink S. (2010). Power Plant Investments in the Turkish Electricity Sector: A Real Options Approach Taking into Account Market Liberalization, FCN Working Paper No. 21/2010, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, December (revised July 2011).
- Melchior T., Madlener R. (2010). Economic Evaluation of IGCC Plants with Hot Gas Cleaning, FCN Working Paper No. 22/2010, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, December.
- Lüschen A., Madlener R. (2010). Economics of Biomass Co-Firing in New Hard Coal Power Plants in Germany, FCN Working Paper No. 23/2010, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, December (revised July 2012).
- Madlener R., Tomm V. (2010). Electricity Consumption of an Ageing Society: Empirical Evidence from a Swiss Household Survey, FCN Working Paper No. 24/2010, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, December.
- Tomm V., Madlener R. (2010). Appliance Endowment and User Behaviour by Age Group: Insights from a Swiss Micro-Survey on Residential Electricity Demand, FCN Working Paper No. 25/2010, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, December.
- Hinrichs H., Madlener R., Pearson P. (2010). Liberalisation of Germany's Electricity System and the Ways Forward of the Unbundling Process: A Historical Perspective and an Outlook, FCN Working Paper No. 26/2010, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, December.
- Achtnicht M. (2010). Do Environmental Benefits Matter? A Choice Experiment Among House Owners in Germany, FCN Working Paper No. 27/2010, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, December.

#### 2009

- Madlener R., Mathar T. (2009). Development Trends and Economics of Concentrating Solar Power Generation Technologies: A Comparative Analysis, FCN Working Paper No. 1/2009, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, November.
- Madlener R., Latz J. (2009). Centralized and Integrated Decentralized Compressed Air Energy Storage for Enhanced Grid Integration of Wind Power, FCN Working Paper No. 2/2009, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, November (revised September 2010).
- Kraemer C., Madlener R. (2009). Using Fuzzy Real Options Valuation for Assessing Investments in NGCC and CCS Energy Conversion Technology, FCN Working Paper No. 3/2009, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, November.
- Westner G., Madlener R. (2009). Development of Cogeneration in Germany: A Dynamic Portfolio Analysis Based on the New Regulatory Framework, FCN Working Paper No. 4/2009, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, November (revised March 2010).
- Westner G., Madlener R. (2009). The Benefit of Regional Diversification of Cogeneration Investments in Europe: A Mean-Variance Portfolio Analysis, FCN Working Paper No. 5/2009, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, November (revised March 2010).
- Lohwasser R., Madlener R. (2009). Simulation of the European Electricity Market and CCS Development with the HECTOR Model, FCN Working Paper No. 6/2009, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, November.
- Lohwasser R., Madlener R. (2009). Impact of CCS on the Economics of Coal-Fired Power Plants Why Investment Costs Do and Efficiency Doesn't Matter, FCN Working Paper No. 7/2009, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, November.
- Holtermann T., Madlener R. (2009). Assessment of the Technological Development and Economic Potential of Photobioreactors, FCN Working Paper No. 8/2009, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, November.
- Ghosh G., Carriazo F. (2009). A Comparison of Three Methods of Estimation in the Context of Spatial Modeling, FCN Working Paper No. 9/2009, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, November.
- Ghosh G., Shortle J. (2009). Water Quality Trading when Nonpoint Pollution Loads are Stochastic, FCN Working Paper No. 10/2009, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, November.
- Ghosh G., Ribaudo M., Shortle J. (2009). Do Baseline Requirements hinder Trades in Water Quality Trading Programs?, FCN Working Paper No. 11/2009, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, November.
- Madlener R., Glensk B., Raymond P. (2009). Investigation of E.ON's Power Generation Assets by Using Mean-Variance Portfolio Analysis, FCN Working Paper No. 12/2009, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, November.

- Madlener R., Gao W., Neustadt I., Zweifel P. (2008). Promoting Renewable Electricity Generation in Imperfect Markets: Price vs. Quantity Policies, FCN Working Paper No. 1/2008, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, July (revised May 2009).
- Madlener R., Wenk C. (2008). Efficient Investment Portfolios for the Swiss Electricity Supply Sector, FCN Working Paper No. 2/2008, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, August.
- Omann I., Kowalski K., Bohunovsky L., Madlener R., Stagl S. (2008). The Influence of Social Preferences on Multi-Criteria Evaluation of Energy Scenarios, FCN Working Paper No. 3/2008, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, August.
- Bernstein R., Madlener R. (2008). The Impact of Disaggregated ICT Capital on Electricity Intensity of Production: Econometric Analysis of Major European Industries, FCN Working Paper No. 4/2008, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, September.

Erber G., Madlener R. (2008). Impact of ICT and Human Skills on the European Financial Intermediation Sector, FCN Working Paper No. 5/2008, Institute for Future Energy Consumer Needs and Behavior, RWTH Aachen University, September.

FCN Working Papers are free of charge. They can mostly be downloaded in pdf format from the FCN / E.ON ERC Website (www.eonerc.rwth-aachen.de/fcn) and the SSRN Website (www.ssrn.com), respectively. Alternatively, they may also be ordered as hardcopies from Ms Sabine Schill (Phone: +49 (0) 241-80 49820, E-mail: <u>post fcn@eonerc.rwth-aachen.de</u>), RWTH Aachen University, Institute for Future Energy Consumer Needs and Behavior (FCN), Chair of Energy Economics and Management (Prof. Dr. Reinhard Madlener), Mathieustrasse 10, 52074 Aachen, Germany.